CREDIT Research Paper



No. 01/19

Foreign Ownership and Wages: Evidence from Five African Countries

by

Dirk Willem te Velde and Oliver Morrissey

Centre for Research in Economic Development and International Trade, University of Nottingham The Centre for Research in Economic Development and International Trade is based in the School of Economics at the University of Nottingham. It aims to promote research in all aspects of economic development and international trade on both a long term and a short term basis. To this end, CREDIT organises seminar series on Development Economics, acts as a point for collaborative research with other UK and overseas institutions and publishes research papers on topics central to its interests. A list of CREDIT Research Papers is given on the final page of this publication.

Authors who wish to submit a paper for publication should send their manuscript to the Editor of the CREDIT Research Papers, Professor M F Bleaney, at:

Centre for Research in Economic Development and International Trade, School of Economics, University of Nottingham, University Park, Nottingham, NG7 2RD, UNITED KINGDOM

Telephone (0115) 951 5620 Fax: (0115) 951 4159

CREDIT Research Papers are distributed free of charge to members of the Centre. Enquiries concerning copies of individual Research Papers or CREDIT membership should be addressed to the CREDIT Secretary at the above address. Papers may also be downloaded from the School of Economics web site at: www.nottingham.ac.uk/economics/research/credit **CREDIT Research Paper**



No. 01/19

Foreign Ownership and Wages: Evidence from Five African Countries

by

Dirk Willem te Velde and Oliver Morrissey

Centre for Research in Economic Development and International Trade, University of Nottingham

The Authors

Dirk Willem te Velde and Oliver Morrissey are both Research Fellows, International Economics Development Group, ODI. Oliver Morrissey is also Director of CREDIT and Reader, School of Economics, University of Nottingham.

Acknowledgements

This is a version of a paper prepared for the UNIDO/CSAE forum on 'New Industrial Realities and Firm Behaviour in Africa', 20-22 september 2001 at St Edmund Hall, Oxford. We thank Simon Appleton, Richard Cornes, Sourafel Girma, Holger Gorg, Andrew McKay, Eric Strobl and Francis Teal for helpful comments and suggestions (even if we were unable to incorporate all of them). We are grateful to DFID for funding under grant R8003. Views and opinions expressed are those of the authors alone. We are also grateful to CSAE for making available the data used.

Foreign Ownership and Wages: Evidence from Five African Countries

by Dirk Willem te Velde and Oliver Morrissey

Abstract

This paper uses data on individual wages in manufacturing industry for five African countries in the early 1990s to test whether firms owned by foreigners pay higher wages than do forms owned by locals for apparently equivalent workers, and whether such benefits accrue to all or only certain types of workers. We present two main findings. First, foreign ownership is associated with a 20-40 per cent increase in individual wages (conditional on age, tenure and education) on average. This is halved to 8-23 per cent if we take into account the fact that foreign-owned firms are larger and locate in high-wage sectors and regions. Secondly, there is a tendency in some countries for more skilled workers (using occupation and education categories) to benefit more from foreign ownership than less skilled workers, and this conclusion holds after accounting for the size distribution of foreign firms. We discuss, but cannot directly test, the plausibility of two explanations for these findings: 1) foreign-owned firms employ technologies that are more skill-biased than technologies in local firms and 2) skilled workers in foreign firms are more effective in rent-sharing than other workers. We contend that these explanations may not be mutually exclusive, hence cannot be empirically distinguished

Outline

- 1. Introduction
- 2. Foreign Ownership and Wages: Existing Evidence
- 3. Foreign Ownership and the Earnings Function
- 4. Data Description
- 5. Econometric Results
- 6. Conclusions and Policy Implications

I. INTRODUCTION

Investment by foreigners is an important source of funds for manufacturing firms in sub-Saharan African (SSA) countries. Foreign presence in SSA would be expected to increase as countries engage in privatisation and as, under the auspices of the WTO, investment is liberalised. Conventionally, the literature on the effects of foreign ownership is concerned with foreign direct investment (FDI). Does FDI contribute to growth, and do affiliates of multinationals behave differently to local firms (e.g. are they more capital-intensive, more productive, and do they pay higher wages)? This literature is reviewed elsewhere (see Velde, 2001) as few of the foreign-owned firms in the sample available to us are actually affiliates of multinationals. In our data foreign-owned is not equivalent to FDI as conventionally understood; in the early 1990s few African manufacturing firms (in the sample countries) were owned (or had significant investment) by foreign companies. In our definition, foreign-owned means some ownership by foreigners, usually individuals that are not nationals although including foreign companies. We will assume that ownership by foreign nationals captures important features of the differences in wage-setting behaviour of foreign and local firms (this is discussed further in the conclusion). This paper focuses on one specific issue, whether firms owned by foreigners (FOR) pay higher wages than do firms owned by locals (LOC) to equivalent workers (i.e. workers with similar levels of experience and education).

The availability of large-scale surveys has supported a noticeable increase in research on manufacturing enterprises in Africa. Much of this was initiated with the World Bank and bilateral donor-sponsored Regional Programme on Enterprise Development (RPED), which has funded surveys in a number of African countries during the 1990s. We use data from the first three waves of surveys (1990-93) for five countries - Cameroon, Ghana, Kenya, Zambia and Zimbabwe (available form the CSAE web-site). A particular advantage of the RPED data is that there are two data sets, one at the firm level with information on firm characteristics, and the other containing data on individuals (wages and employee characteristics). Our primary concern is with the information on individual employees, and the firm-level data is used to identify the characteristics of the firms in which they are employed (in particular whether there is foreign ownership). We

2

examine whether ownership by foreigners is associated with higher wages distinguishing between types of workers.

A number of recent studies analyse these surveys (e.g. Bigsten et al. 1997, 1999, 2000; Strobl and Thornton, 2001; Teal, 2000; Söderborn and Teal, 2001a and 2001b), but most are primarily concerned with firm-level data and none specifically address the questions raised in this paper. Some of these studies do include foreign ownership as a control variable. Bigsten et al. (2000) examine rates of return on physical and human capital and include results for earnings functions similar to those we estimate (we therefore check our results against theirs). Whilst they include ownership type as a control variable, they did not publish the results for this. Strobl and Thornton (2001), using the same crosscountry data set that we use, do find that FOR tend to be larger and pay higher wages, as does Mazumdar (1995). However, theses studies are concerned with explaining the average wage paid by firms, and specifically with the employer size-wage effect, whereas we are concerned with factors influencing the wage received by workers. Söderbom and Teal (2001a: 18) find foreign ownership to be insignificant in estimating a Cobb-Douglas production function for Ghanaian firms, i.e. ownership does not have an independent effect on output, but are not concerned with earnings at the employee level. Where appropriate, we will refer to evidence from these other studies when interpreting our results.

The question of *which* type of worker gains from foreign ownership has largely been ignored in the literature. There are some initial, but crude, results showing that non-production workers benefit more from foreign ownership than production workers in terms of wages, see Lipsey and Sjöholm (2001) for Indonesia, Aitken *et al.* (1996) for Mexico and Venezuela, Feenstra and Hanson (1995) for Mexico. This paper tests whether foreign firms pay *all* of their workers more than similar workers in local firms, or whether a foreign ownership wage premium pertains to more skilled and educated workers only. Such evidence would shed new light on the potential direct effects of foreign ownership on low-income groups, as low-income workers are usually less skilled and less educated workers. As Strobl and Thornton (2001) observe, if larger firms tend to pay higher wages (and the same applies for FOR) this may contribute to (wage) inequality. However, this is only part of the full picture of the long-run effects of foreign ownership on income and employment of low-income workers, and we discuss this in

the paper. The results can also be used as a first step to assess whether foreign-owned firms employ technologies that are more skill-biased than local firms and/or whether skill workers in foreign-owned firms are more successful in rent-sharing than less skilled workers. We return to these issues in the conclusion.

Much of the evidence on foreign ownership and wages relates to the firm level, and not to the level of individuals. Consequently, it is difficult to assess which types of workers gain from foreign ownership. It may be observed that larger firms and/or FOR pay higher wages (on average) and employ a greater proportion of skilled workers, but this does not tell us whether an employee with given characteristics will earn more if employed by an FOR than by a local firm, for example. Furthermore, with firm level data alone it is difficult to distinguish between the contribution of the mix of worker characteristics to the average wage as against the effect of firm characteristics on the wage paid to an average worker (and to wage differentials among workers). Thus, for example, Strobl and Thornton (2001) find that the 'raw' size-wage differential is about 60% but this falls to 20-30% if one controls for observable worker and firm characteristics. They therefore find that large firms pay higher wages even controlling for other factors. We can ask the complementary question, do workers with given characteristics earn higher wages in large firms (and in FOR). Taking the two pieces of evidence together can shed more light on the nature of wage differentials in SSA countries.

The structure of the paper is as follows. Section 2 presents a brief review of theory and evidence on the effects of foreign ownership on wages. Section 3 then presents the wage determination model used, essentially a Mincerian framework. Section 4 discusses the data and relevant results from previous studies using this database. Section 5 presents our results, assessing whether foreign-ownership benefits specific type of workers. Conclusions are presented in Section with a discussion of alternative explanations for our results and the policy implications.

II. FOREIGN OWNERSHIP AND WAGES: EXISTING EVIDENCE

There is a considerable literature on the complex relationships between foreign ownership and wages, mostly concerned with affiliates of foreign firms rather than firms owned by foreign nationals. In both types of foreign ownership the difference with locally owned firms is greater access to (foreign) capital, but whereas multinationals should also bring access to technology this is not obviously the case for foreign individuals. While greater access to capital implies a greater ability to import capital goods, and this is a determinant of skill-biased technical change, foreign individuals need not be as likely to do this as affiliates of foreign companies (the latter will have an incentive to import from the parent). This distinction is important when it comes to interpreting our results against alternative explanations for wage premium effects in the literature.

The literature on multinationals suggests that the presence of a firm-specific asset explains in part the observation of a wage differential between foreign-owned and local firms (Dunning, 1993). Affiliates of multinationals use more up-to-date technologies, require more skilled workers, have access to better inputs, are more productive and hence can pay more. Firms owned by foreign individuals may not exhibit these characteristics. However, there could also be other reasons for a wage differential. For instance, it is often argued that foreign firms want to attract the best workers with the highest ability and work effort and as a consequence need to pay more. This is linked to the efficiency-wage argument to pay workers more in order to raise effort and productivity. Another explanation could be that foreign-owned firms sort on education, in the absence of knowledge of the quality of local workers, and will pay more accordingly (this argument would apply to foreign nationals as well as to multinationals). Finally, foreign firms may be more profitable than local firms and, as Blanchflower *et al.* (1996) argue, wages can be positively correlated to profits, which is shown empirically in the case of Ghana (Söderborn and Teal, 2001a). We will return to these different hypotheses later.

Observing a correlation between foreign ownership and wages does not necessarily imply a causal link. To determine the effects of foreign ownership on wages, one should control for other determinants of wages. If we fail to control for such other determinants we ascribe effects on wages to foreign ownership, whereas the effects could be due to underlying factors which happen to be more prevalent in foreign-owned firms. When examining the relationship between foreign ownership and wages at the firm level (i.e. why average wages vary by firm type), the following control variables are the most important: skill intensity, size, location and sector dummies.¹

<u>Skills</u> Foreign-owned firms tend to have better access to best-practice technology, requiring more skills to operate complex production processes. A higher skill mix explains in part a higher average wage bill for employees in foreign-owned firms. Strobl and Thornton (2001) find that worker characteristics explain a large part of the employer size-wage effect in SSA countries. The different skill mix explains differences in average firm wages.

<u>Size</u> Foreign-owned firms are usually relatively large. Large firms pay more for observationally equivalent workers than small firms do, and also tend to hire more educated workers, particularly white-collar workers, possibly because greater monitoring problems in large firms require more productive managerial workers (Polachek and Siebert, 1993: 261). This different skill mix does not explain fully why large firms pay higher average wages. Strobl and Thornton (2001) show that there is a substantial size-wage premium (controlling for observable worker characteristics, large firms pay 20-30% higher wages in the five SSA countries), which is higher for non-production workers. Thus, it is important to control for firm size.

<u>Industry</u> Foreign firms may locate in particular industries. If there are large and persistent wage differentials across industries and FOR locate in higher paying industries, this will give rise to an apparent tendency for FOR to pay higher wages.

<u>Location</u> It is also important to control for location of operation. Multinationals (at least in manufacturing) tend to locate in areas where there is sufficient high-skill labour, which are often high-wage locations. There is also a tendency to agglomerate in clusters (often close to the capital city) and while this reduces firm costs in terms of learning it also pushes up local wages (e.g. Fujita *et al*, 2000). The impact of regional controls will be more important the less competitive and integrated labour markets are across the spatial dimension. Strobl and Thornton (2001) find that firms located in the capital city tend to be larger and to pay higher wages.

¹ Other controls in the wage functions include union density, share of female workers, share of foreign workers and, less often, the firm age.

Why are some types of workers paid more?

The evidence cited above relates to the firm level, and not to the level of individuals; i.e. the controls are firm characteristics that explain higher average wages at a firm level. An ownership-wage premium at the firm level may not translate into the same wage premium for all types of workers within FOR, or within firms having any specific characteristic. Our concern is whether firm characteristics affect the wage paid to workers with the same observable characteristics, in particular if FOR pay higher wages to equivalent workers. We consider three explanations for why FOR may pay higher wages - bargaining, technology and management.

The <u>bargaining explanation</u> is based on the hypothesis that skilled workers are in a stronger bargaining position than less-skilled workers and hence are more effective in rent-sharing (i.e. can capture a greater share of profits). This can be because skilled workers are in relatively scarce supply, or simply because they are more effective in the bargaining process, e.g. due to better negotiation skills or a louder voice. While bargaining power helps to explain why skilled workers earn higher wages (than would be explained simply by their productivity), it does not explain why skilled workers in one type of firm (e.g. large or FOR) may be paid more than skilled workers in the skill premium by type of firm.

Söderbom and Teal (2001a) find that size is the single most important determinant of earnings, controlling for fixed firm effects, and try to test between rent-sharing and efficiency wage (risk-sharing) explanations. They find no evidence to support the efficiency wage hypothesis (earnings are not a predictor of productivity). However, evidence that predicted output and lagged profits do influence earnings is cited in support of the rent-sharing argument.² Thus, skilled workers earn higher wages in larger firms because of their bargaining power. However, controlling for the size effect, why

² While convincing, the tests are not conclusive as it is difficult to distinguish the case where firms that pay higher wages are more productive or profitable (efficiency wages) from more profitable and productive firms being able to pay higher wages (rent-sharing).

should skilled workers have greater bargaining power in FOR as against LOC? Skilled workers may have relatively stronger bargaining power in FOR that want to attract relatively scarce skilled workers but lack detailed information on the local labour market. A number of surveys (UNCTAD, 2000a; Business Map, 2000) indicate that the availability of skills is one of the major barriers to investment in African countries. Thus, FOR are more likely to sort workers on the basis of observable skill characteristics, and this would give rise to rent sharing. Foreign-owned firms that are less familiar with the (availability of) skills in the local labour market are more likely to sort on observable worker characteristics when recruiting new employees.

The <u>technology explanation</u>, by contrast, is based on the idea that certain types of firms (larger or FOR) pay higher wages to skilled workers because such workers are more productive (after controlling for observable worker characteristics), while less-skilled workers have the same productivity in all firms. The basic argument is that some firms are more capital-intensive and/or use more advanced technologies, hence are more skill-intensive. In particular, FOR are expected to be more capital intensive. This need not be because they are more efficient than local firms. Rather, it is because FOR face a lower cost of capital and therefore use relatively more capital (and skill) intensive techniques. Söderbom and Teal (2001b) do not find that foreign firms are more efficient than local firms in Ghana, hence we cannot assume that relative productivity explains the wage differential (for FOR as against LOC). Technology can explain why FOFR employ relatively more skilled labour, but if it is to also explain why they are paid more one would have to show that skilled labour is more productive in FOFs. Görg and Strobl (2001) find that imports of capital goods (technology) explain skill-biased technological change in Ghana, but this relates to firm-level wage differentials.

Within this technology explanation, we need to account for the stylised fact that foreign firms are larger than local firms, as the employer-size wage premium appears to be larger for non-production than for production workers. We may find that foreign firms pay higher wages to more-skilled workers because they are larger, not because they are foreign-owned. Strobl and Thornton (2001) review various options and find that the technology explanation for the differential size-wage premium is the most promising. Using a special dataset for Ghana they find that the size premium varies by skill only in technology intensive firms. To the extent that foreign firms are more technology

intensive and produce more complex products than local firms, we would expect that foreign firms are associated with higher wage differentials for more skilled workers, over and above the effect on the wage differential by skill coming from size effects alone. We can test for this by accounting fully for a differential size-wage premium depending on employee's occupation.

The <u>management explanation</u> for why foreign-owned firms pay higher wages to skilled workers, but not to less-skilled workers, is that management and organisational techniques attach more importance to skilled workers. Foreign firms may be more prone to organisational change, especially when foreign firms acquire (or merge with) local firms, and such change can be more effectively managed by skilled workers. In addition, new management techniques introduced into a country by foreign-owned firms (e.g. justin-time or lean management techniques), attach more importance to individuals operating at a lower level but who need considerable skills to be successful. The management explanation implies that skilled workers in foreign firms are more productive than in local firms, and is observationally equivalent to the technology explanation. In this paper, we treat the management explanation as part of the technology explanation.

III. FOREIGN OWNERSHIP AND THE EARNINGS FUNCTION

We use and extend the framework of Mincer (1974) to examine the effects of foreign ownership on wages of individuals. This basic framework has been applied by Bigsten *et al.* (2000) to the database we use. The starting point is to estimate the following equation:

$$\log(Y_{it}) = \alpha + \sum_{j} r_j S_{ij} + \beta_1 age_{it} + \beta_2 age_{it}^2 + \gamma_1 ten_{it} + \gamma_2 ten_{it}^2 + \varepsilon_{it}$$
(1)

 Y_{it} is a measure of the wage of individual i=1,...,N at time t=1,...,T. S_{ij} is a 0/1 dummy which is 1 for the highest level j of education completed (or number of years of schooling in the original Mincerian framework) – we include all levels of education except the first (no education), hence j=1,...,J-1, and r_j is the rate of return to the completion of education level j. Experience is captured by employee's *age* and *ten*, the number of years employed by the current firm (tenure), and the squared terms all for

non-linear effects. Appendix A discusses some issues concerning this type of equation, in particular concerning the underlying theory and estimating coefficient r_j . The substance of this paper is to include foreign ownership in (1) in a number of ways to assess which types of workers benefit from foreign ownership.

The first extension is to include a 0/1 dummy FOR_i denoting whether the firm in which individual *i* is employed has any foreign ownership:

$$\log(Y_{it}) = \alpha + \sum_{j} r_j S_{ij} + \beta_1 age_{it} + \beta_2 age_{it}^2 + \gamma_1 ten_{it} + \gamma_2 ten_{it}^2 + \varphi FOR_i + \varepsilon_{it}$$
(2)

The coefficient φ is the percentage increase in wages enjoyed by individual *i* because s/he is employed in a foreign-owned firm. However, as discussed, the coefficient φ may overstate the true effects if foreign ownership is correlated with control variables (Z_k , the firm characteristics such as size, sector, etc.) that are positively correlated with the dependent variable. Equation (3) includes control variables (k=1,...,K).

$$\log(Y_{it}) = \alpha + \sum_{j} r_j S_{ij} + \beta_1 age_{it} + \beta_2 age_{it}^2 + \gamma_1 ten_{it} + \gamma_2 ten_{it}^2 + \varphi FOR_i + \sum_k \zeta_k Z_{ik} + \varepsilon_{it}$$
(3)

We then estimate (4) to assess whether the mark-up in wages (= φ in (3)) occurs for workers in all sectors (control variables $Z_{SEC,l}$ equal 1 for sector l=1,...,L and 0 otherwise), or workers in some sectors only:

$$\log(Y_{it}) = \alpha + \sum_{j} r_j S_{ij} + \beta_1 age_{it} + \beta_2 age_{it}^2 + \gamma_1 ten_{it} + \gamma_2 ten_{it}^2 + \sum_{l} \varphi_l FOR_i Z_{i,SEC,l} + \sum_{k} \zeta_k Z_{ik} + \varepsilon_{it}$$

$$(4)$$

Regression equation (5) estimates (3), but interacts the variable *FOR* with education level *S* (here for j=1,...,J) to assess whether foreign ownership is beneficial for individuals regardless of the level of education completed.

$$\log(Y_{it}) = \alpha + \sum_{\substack{j=1,\dots,J-1}} r_j S_{ij} + \beta_1 age_{it} + \beta_2 age_{it}^2 + \gamma_1 ten_{it} + \gamma_2 ten_{it}^2 + \sum_{j=1,\dots,J} \varphi_j FOR_i S_{ij} + \sum_k \zeta_k Z_{ik} + \varepsilon_{it}$$
(5)

Foreign ownership is more likely to have a direct and positive effect on poverty if φ_j is positive and significant for individuals with the lowest levels of education. Given the evidence discussed in Section 2, we would expect φ_j to be higher with higher levels of education. Finally, (6) repeats (5) but replaces levels of education with types of occupation (*OCCUP*) to assess whether foreign ownership affects individuals equally regardless of the type of occupation. Given the evidence and theory discussed earlier, we would expect φ_j in (6) to be higher with more complex and skilled occupations.

$$\log(Y_{it}) = \alpha + \sum_{\substack{j=1,\dots,J-1}} r_j OCCUP_{ij} + \beta_1 age + \beta_2 age^2 + \gamma_1 ten + \gamma_2 ten^2 + \sum_{\substack{j=1,\dots,J}} \varphi_j FOR_i OCCUP_{ij} + \sum_k \zeta_k Z_k + \varepsilon_{it}$$
(6)

When using the interaction terms between foreign ownership and occupation/education we assume that observable worker and firm characteristics are the only determinants of worker wages. If this is not so, for instance if unobservable worker or firm characteristics affect wages, the φ_j coefficients will be biased if foreign ownership is correlated with the unobservables. One can allow for firm specific effects by first differencing and availing of the panel nature of the firm-level data (e.g. Söderbom and Teal, 2001a). However, it is not possible to allow for worker specific effects as we have data on a repeated cross-section basis and not a panel for individual employees. The same firms are interviewed over time, but the workers interviewed within these firms are not necessarily the same.³

IV. DATA DESCRIPTION

The data in this paper draw from firm-level surveys in Cameroon, Ghana, Kenya, Zambia and Zimbabwe as part of the Regional Programme on Enterprise Development

³ Strobl and Thornton (2001) note that as some workers are interviewed more than once, there is a potential for correlated errors if multiple observations for the same worker are included. They try to test for this with the Ghanaian data and find no evidence for a bias in results.

(RPED) conducted in repeated waves during the 1990s. In the data set we use (that available on the CSAE web-site) there are three years (waves) of data for most of the five countries, covering firms in four manufacturing sectors: food, wood, textiles and metal. The dataset includes formal and informal firms of various sizes, and is thought to be representative of the manufacturing sector in the respective countries.

We link two data sets, one containing data on firm characteristics (RPED), such as location, sector, ownership structure and another containing data on individuals (EARN), such as education, occupation, tenure, age and wages. The two databases can be linked through a country specific firm identifier in addition to data on waves. The data relate to two or three different years, and as there are insufficient time series, we pooled data across waves and focus on a static framework.

We use the variable 'any foreign ownership - ANYFOR ' as a 0/1 dummy to define if a firm is FOR. There are different types of wage data. We use monthly earnings data in current domestic prices and also in current dollars converted using PPP values (see Bigsten *et al.*, 2000, for the PPP's used). The dependent variable in the regression analysis is in logs. An important part of the analysis in this paper relates to the education and occupation data. The data distinguish between five different levels of education: no education, some primary education, primary education completed, secondary education completed and university. Of course, quality of the various levels of education may not be comparable, so that we need to exercise caution in interpreting regression results that use data pooled across countries. There are also various types of occupations: management, administration, sales staff, supervisor, technician and production worker, and for some countries apprentice and/or master. We take the first four of these occupations (and master) as 'skilled' and the rest as 'unskilled', as does Teal (2000).

The Appendix Table A1 summarises information on key variables by country. It reports the mean, distinguishes between three waves and between two types of ownership: local firms (LOC) and foreign-owned firms (FOR). Individuals in foreign-owned firms appear to have higher wages, longer tenure within the present firm, are better educated, are older and are employed in firms that are larger. These are well-known facts about differences between foreign-owned and local firms in developing countries, as discussed at great length in Dunning (1993). The differences exist for all five countries, but it would be interesting to test whether these differences are also significant.

Table 1 Influences on Likelihood of Foreign Ownership	Table 1	Influences on	Likelihood of I	Foreign ()wnership
---	---------	---------------	------------------------	-----------	-----------

	Logit regression
Cameroon	1.24
	(5.3) *
Ghana	0.58
W.	(2.9) *
Kenya	0.20
7	(1.0)
Zambia	-0.11
Located in capital city	(-0.5) 0.12
Located in capital city	(0.9)
Wood and furniture sector	-0.02
wood and furniture sector	(-0.1)
Textile sector	0.35
	(1.7)
Metal sector	0.85
	(5.5) *
Log (employment)	0.68
	(14.2) *
Share of non-production workers	0.78
	(2.1) *
Constant	-5.0
	(-15.8) *
N	20/0
N Mariana Lihaliha d	2060
Maximum Likelihood	-825.1

Dependent variable = 1 if FOR, 0 otherwise

Notes: Zimbabwe is the omitted country, Food the omitted sector. White heteroscedasticity-consistent *t*-statistics in parenthesis; * indicates significance at least at the 5% level.

Table 1 presents the results of a simple logit estimation to see which characteristics are significantly correlated with foreign ownership in the pooled sample (i.e. pooling all five countries). Allowing for country and sector effects, larger firms and firms with a greater share of non-production workers are more likely to have foreign ownership. Firms in Cameroon and Ghana (relative to Zimbabwe) are more likely to have foreign ownership,

as are firms in the metal sector (relative to food). Location in the capital city does not appear to increase the probability of the firm having foreign ownership.

It is interesting to see which type of non-production worker is more prevalent in foreign firms. Chart 1 summarises the results of the following SUR estimation, where s_{ki} = employment share of occupation group k^4 in firm *i* and *X* includes relevant controls such as size, location, and country and sector dummies:

$$s_{ki} = \alpha_k + \beta_k FOR + \gamma_k' X_{ki} + \varepsilon_{ki}$$

The parameter of interest, β_k , reflects the difference in the proportions of occupation groups employed in foreign and local establishments. We can assume that errors ε_{ki} are correlated across groups (hence SUR estimation). The results indicate that foreign firms employ significantly more managers and specialised staff, but significantly fewer production workers. This is indeed one of the reasons why foreign firms are likely to pay higher wages on average.



Chart 1Foreign Ownership and Employment Structure of Firms by Occupation

(percentage point differences in shares between foreign and local establishments = β)

Thin lines delineate 95 per cent confidence interval around estimated values

⁴ The occupation group TECH has been omitted due to singularity in the SUR system.

Before presenting the results, it is helpful to be clear on the issues that we do not address (due to data constraints):

- Our results relate to firms with any foreign ownership; affiliates of multinationals (FDI) are too few in the sample for meaningful analysis.
- As the employee data is not a panel, we cannot account for unobserved workerspecific effects, nor are we able to account for any possible bias from having some multiple observations for the same worker.
- As we are concerned with earnings at the individual level, we cannot account for unobserved firm-specific effects nor are we able to consider skill-based wage differentials at the firm level.
- We cannot comment on any long-run impact of foreign ownership on wages, as the data cover at best three years over 1990-93.
- We cannot comment on the overall effect on employment, as we do not have information about manufacturing as a whole in each country.
- We cannot comment on effects of foreign firms on the level of wages in an economy or sector (we do not have data on the share of foreign employment in a sector as a whole, which is typically used in spillover studies).

V. ECONOMETRIC RESULTS

In this section we discuss the results of estimation of equations (1) to (6) (and others) for manufacturing in the five African countries. We began by estimating (1) for each country to check whether our results are consistent with those reported in table 4 of the working paper version of Bigsten *et al.* (2000).⁵ The results are summarised in Table 2 (and presented in detail in Appendix A) and are the same except for minor differences, such as values of *t*-statistics, which may be due to the use of a different statistical package.⁶ Most coefficients are well determined and consistent with expectations.⁷ We can be reasonably confident that this framework is appropriate for testing effects of foreign ownership.

The effects of foreign ownership are positive, substantial and significant in all regressions (Table 2). The first row gives the raw 'FOR premium' without controls.

⁵ The published version collapses various levels of education into one education variable.

⁶ We us STATA. The *t*-statistics are computed using heteroscedastic consistent errors (White, 1980), and hence we do not report tests for heteroscedasticity.

Wages for individuals in foreign-owned firms are 20 per cent higher in Cameroon, 24 per cent in Kenya, 30 per cent in Zimbabwe, and 37 per cent in Ghana and Zambia. For most countries the inclusion of *FOR* does not alter the significance level of other variables in specifications (1) and (2), and has only a minimal effect on the values of some coefficients (Appendix Tables A2-A6). In the case of Ghana (Table A3), the coefficient on *tenure* becomes insignificant when *FOR* is included, suggesting that it is only long tenure in a foreign-owned firm, rather than tenure *per se*, that increases earnings. Employees of firms located in the capital city receive a wage premium of some 12-28%, controlling for other factors (Appendix A).

Table 2	le 2 Summary of Main Results from Wage Equations					
	Cameroon	Ghana	Kenya	Zambia	Zimbabwe	
'FOR premium'						
No controls	0.20	0.37	0.24	0.37	0.30	
Firm controls	0.08	0.22	0.17	0.23	0.13	
Which						
workers?						
Sectors	ns	ns	Wood	Metal	Food	
			Food	Food		
Education	SECC	PRIMC	SECC	PRIMN	PRIMC	
		SECC		SECC	SECC	
		UNIVC				
Occupation	MGMT	ns	All but	non-Prod	ns	
	ADMIN		Sales	(all)		
	SALES					
Size premium						
With controls	0.13	0.15	0.09	0.10	0.14	
ST estimates	0.23	0.27	0.27	0.21	0.39	

Notes: Summary of results from Tables A2-A6; ns implies non-significant differences in coefficients (based on P-values). '*FOR* premium' is coefficient on *FOR* in specifications (2), no controls, and (3), with firm specific controls. Sectors are the significant interactive (*FOR**sector) terms in specification (4). Education are the significant interactive (*FOR**education) terms in specification (5). Occupation gives the significant interactive (*FOR**occupation) categories in specification (6). Size premium is the coefficient on log(employment) in specification (3),

⁷ We are not interested in the level of the coefficient for education, see appendix A.

Tables A2-A6. The final column gives estimates of the size-wage effect for all workers with all controls from Strobl and Thornton (2001: Table 3).

The effects of foreign ownership on earnings are substantially reduced when taking firmspecific control variables into account (the second row of Table 2). The results nevertheless suggest that individuals in foreign-owned firms earn higher wages, the premium ranging from eight per cent in Cameroon, 13 per cent in Zimbabwe, 17 per cent in Kenya, 22 per cent in Ghana, to 23 per cent in Zambia. Firm size, as expected, is significant for all countries. Controlling for the other variables, wages of employees in larger firms are higher by some 9-15% (the size-wage effect is smallest in Kenya and greatest in Ghana). Controlling for foreign ownership reduces significantly the sizewage premium estimated by Strobl and Thornton (the final two rows of Table 2) We show that the FOR premium applies fairly uniformly even when one accounts for firm size.

The premium associated with foreign-ownership applies to all workers, but tends to be more pronounced for skilled and non-production workers. However, these skill or occupation-specific results do vary by country, often quite significantly, and are the focus of the remainder of this section. Nevertheless, the most senior workers tend to benefit the most from being employed in foreign-owned firms. The FOR premium is also associated with the educational attainment of workers. In all countries, workers that have completed secondary education benefit from being employed in FOR rather than LOC. In some countries, workers with primary education also gain a FOR premium, but this is not a strong consistent result. There are also some sector effects, although these are difficult to interpret: there appears to be a premium to being employed in foreign-owned firms in the Food sector in Kenya, Zambia and Zimbabwe.

What type of worker benefits from foreign ownership?

To assess if the wage 'premium' from foreign ownership applies equally to different types of worker, we estimate specifications (5) and (6), which omit the *FOR* term. Given that there is a FOR premium, these specifications estimate how it is distributed by educational or occupational level. The results for (5) are presented in Table 3. There are some interesting patterns. Earnings do appear to increase with educational attainment.

The premium on completing primary education (PRIMC) varies from 12% to 37% where significant, whereas that on completing secondary education (SECC) varies from 31% to 81% and is always significant. The premium on a University education appears to be high. Individuals who have not completed primary education (either NONE or PRIMN) do not enjoy a wage premium in foreign-owned firms (the exception is Zambia but the P-value indicates that the coefficients are not significantly different). On the other hand, individuals who have completed secondary education benefit significantly from foreign ownership in all countries under consideration, with an implied premium ranging from 16 per cent in Cameroon to 33 per cent in Kenya.

	Cameroon	Ghana	Kenya	Zambia	Zimbabwe
PRIMC	0.16	0.06	0.14	0.37	0.12
	(2.9) *	(0.9)	(4.8) *	(5.9) *	(2.5) *
SECC	0.53	0.31	0.35	0.81	0.53
	(9.1) *	(3.9) *	(10.6) *	(11.2) *	(8.4) *
UNIVC	1.31	1.44	1.28	1.82	1.73
	(15.5) *	(10.7) *	(10.3) *	(20.8) *	(20.5) *
Foreign ownership (FOR)					
Log (employment)	0.13	0.15	0.09	0.11	0.14
	(8.8) *	(9.1) *	(10.7) *	(6.1) *	(11.9) *
FOR * NONE	0.02	0.14		0.31	-0.08
	(0.1)	(0.1)		(0.9)	(-0.8)
FOR * PRIMN	-0.02	-	-0.05	0.25	0.06
	(-0.1)		(-1.1)	(2.0) *	(0.8)
FOR * PRIMC	-0.02	0.24	0.06	0.10	0.12
	(-0.4)	(6.6) *	(1.6)	(1.4)	(2.4) *
FOR * SECC	0.16	0.21	0.33	0.33	0.23
	(3.2) *	(3.2) *	(6.9) *	(4.3) *	(3.2) *
FOR * UNIVC	0.15	-0.38	0.37	0.40	-0.66
	(1.3)	(-2.0) *	(1.7)	(2.3)	(-2.3)
Time Dummies	Yes	Yes	Yes	Yes	Yes
Sector Dummies	Yes	Yes	Yes	Yes	Yes
Ν	1534	2257	3035	1593	1866
R-squared	0.52	0.51	0.39	0.48	0.39
Test	P=0.07	P=0.00	P=0.00	P=0.16	P=0.004

Table 3Wage Equations with Education Attainment, specification (5)

Notes: Dependent variable is log of monthly wages in current domestic currency. White heteroscedasticityconsistent *t*-statistics in parenthesis; * indicates significance at least at the 5% level. Equation as specified in the text. The P-values are the probability level that coefficients on FOR*education interaction terms are significantly different. Standard worker controls (age, tenure, male, etc.) are included and reported in Appendix tables, which also report coefficients on firm location, state ownership.

For those countries where there are significant differences between coefficients on $FOR*S_j$, i.e. Ghana, Kenya and Zimbabwe, the latter two show a clear tendency for individuals with a higher level of education completed to benefit more from foreign ownership, at least for the first four levels of education identified. As there are relatively few observations for university educated individuals in foreign owned firms, we attach less weight to those observations. Nonetheless, individuals with low levels of education do not lose as a result of being employed in foreign-owned firms as none of the coefficients on low levels of education are estimated to be significantly negative.

Table 4 presents the results for the estimation of (6) using occupation rather than education dummies. As expected, managers earn more than most other types of occupations, followed by administrators, sales and supervisors, who earn more than technicians and production workers (the base line category). This can be partly explained by the fact that such occupations require more education and skills. Examining the interaction terms between *FOR* and the various occupations, there is a clear tendency for more senior occupations (managers and administrators) in almost all countries, and relatively skilled jobs - supervisors, sales and technicians (in some cases) - to benefit more from being employed in foreign owned firms. Production workers, except in Kenya, do not benefit from being employed by foreign-owned firms. These effects are significant only in those countries for which the null hypothesis of equal coefficients is rejected (see P-values), i.e. Cameroon, Kenya and Zambia. Other coefficients are consistent with earlier results.

l able 4	Wage Equations with Occupation, specification (6).						
	Cameroon	Ghana	Kenya	Zambia	Zimbabwe		
MGMT	0.87	0.98	0.84	1.19	1.23		
	(8.9) *	(11.9) *	(6.4) *	(13.6) *	(9.8) *		
ADMIN	0.34	0.37	0.45	0.52	1.41		
	(5.1) *	(7.0) *	(6.7) *	(6.9) *	(9.2) *		
SALES	0.22	0.27	0.30	0.36	0.62		
	(4.1) *	(4.5) *	(3.8) *	(5.3) *	(10.6) *		
SUPER	0.25	0.39	0.31	0.25	0.41		
	(3.0) *	(7.0) *	(6.7) * $(6.9) *$ 0.30 0.36 $(3.8) *$ $(5.3) *$ 0.31 0.25 $(7.0) *$ $(4.3) *$ 0.06 0.16 (1.9) $(3.0) *$ 0.11 0.17 $(10.1) *$ $(10.3) *$ 0.69 0.73 $(2.0) *$ $(5.0) *$ 0.49 0.30 $(4.1) *$ $(2.1) *$ -0.21 0.25 (-1.2) $(2.2) *$ 0.47 0.35 $(4.4) *$ $(3.3) *$	(9.4) *			
TECH	-0.09	0.09	0.06	0.16	0.41		
	(-1.5)	(1.8)	(1.9)	(3.0) *	(7.0) *		
MASTER	-0.36	0.11					
	(-7.8) *	(2.1) *					
APPRENTICE	-1.30	-1.36			0.24		
	(-2.3) *	(-23.9) *			(1.1)		
Log (employment)	0.17	0.11	0.11	0.17	0.16		
	(10.9) *	(8.0) *	(10.1) *	(10.3) *	(14.0) *		
FOR * MGMT	0.35	0.18	0.69	0.73	0.47		
	(2.5) *	(1.6)	(2.0) *	(5.0) *	(2.5) *		
FOR * ADMIN	0.25	0.15	0.49	0.30	-0.32		
	(2.8) *	(2.1) *	(4.1) *	(2.1) *	(-0.8)		
FOR * SALES	0.14	0.09	-0.21	0.25	0.09		
	(2.0) *	(1.1)	(-1.2)	(2.2) *	(1.2)		
FOR * SUPER	0.15	0.14	0.47	0.35	0.17		
	(1.2)	(1.8)	(4.4) *	(3.3) *	(2.0) *		
FOR * TECH	0.09	0.24	0.35	0.25	0.26		
	(1.2)	(3.3) *	(3.2) *	(2.0) *	(1.6)		
FOR * PROD	-0.02	0.04	0.15	-0.00	0.02		
	(-0.4)	(1.1)	(3.4) *	(-0.1)	(0.5)		
FOR * MASTER	0.11	0.03					
	(0.5)	(0.2)					
FOR * APPRENTICE		1.11			0.14		
		(10.9) *			(0.6)		

Table 4 Wage Equations with Occupation,	specification (6).
---	--------------------

Ν	1534	2257	1937	1593	1866
R-squared	0.48	0.68	0.41	0.50	0.50
Test	P=0.02	P=0.12	P=0.00	P=0.00	P=0.08

Notes: As for Table 3. Time and Sector dummies included.

Foreign ownership, size and varying wage premia by skill level

We now consider if the foreign ownership wage premium varies by skill level after taking into account that FOR tend to be large firms and the employer-size premium varies by skill level. Are senior and more skilled workers paid more in foreign-owned firms simply because such firms tend to be larger? We offer two pieces of evidence to suggest that this is not the case, i.e. controlling for size, relatively skilled workers earn a higher wage in *FOR*.. The first is a non-parametric regression on pooled data, and the second is a parametric regression for each country individually. We first estimate:

$$\log(Y_{itc}) = \alpha_c + \sum_j r_j S_{ij} + \beta_1 age + \beta_2 age^2 + \gamma_1 tenure + \gamma_2 tenure^2 + \sum_k \zeta_k Z_k + \varepsilon_{it}$$
(7)

This is a variant of (2), excluding the *FOR* term and estimated on data pooled across all countries. The estimation includes fixed effects for four countries (c = Ghana, Kenya, Zambia and Zimbabwe), sectors, country-specific waves, dummies for male workers, location and state ownership. We save estimated residuals u_{it} from (7) and, in the second stage, regress u_{it} non-parametrically on the log of employment for two levels of skill (skilled and unskilled workers), and by ownership (FOR and LOC). Chart 2 contains the results of the Nadaraya-Watson kernel estimator with bandwidth 0.75 (approximation of true bandwidth that minimises the sum of errors), using a Gaussian kernel function for a 100-point grid (see Blundell and Duncan, 1998).

Chart 2 Kernel estimates of size-wage distributions

by skill of employee and ownership of firm, whole sample.



There are three important results. First, it confirms the existence of an employer-size pay premium (as found parametrically in Strobl and Thornton, 2001). Secondly, foreign firms pay a higher wage than local firms along the whole size distribution (except for some very large firms, but numbers of firms are low at this tail of the distribution). Finally, the pay-premium appears to be greater for skilled workers (and is fairly uniform by firm size), although the difference may not be significant; in general, unskilled workers also receive higher wages if employed in FOR. The tendency of larger firms to provide a pay premium predominantly for skilled workers is often explained by the introduction of skilled-biased technology in such firms (Görg and Strobl, 2001; Strobl and Thornton, 2001).⁸ The evidence presented here is consistent with both the hypothesis that foreign firms introduce technologies that are more skill-biased than local firms, after accounting for the size distribution, and the hypothesis that skilled workers in foreign firms are more effective in rent-sharing.

Using parametric regressions, Table 5 includes an interaction term between size and skill. The significant coefficients suggest that larger firms pay higher wages at all skill levels, but the differential is greater for skilled workers (except for Cameroon, the differential is twice as great for skilled workers). Controlling for this, the significant and differing wage premia found for Cameroon, Kenya and Zambia in Table 4 still apply. The FOR wage differential for these countries (and Ghana) is also *significantly* different for different skill levels as indicated by the P-values at the bottom of each column. In Cameroon and Zambia, it appears that only skilled workers benefit from being employed in FOR, whereas for Kenya all workers benefit (but skilled by about twice as much). The results for Ghana are unusual, as it appears that only unskilled workers receive a FOR wage premium. However, in Table 4 the coefficients on *FOR**occupation for Ghana were not significantly different from each other. It does appear that apprentices in

⁸ Görg and Strobl (2001) do not actually include firm size, or foreign ownership, as determinants of the skilled/unskilled wage premium they estimate. However, they find that imports of foreign capital are a significant determinant of the skill premium, offering support for skill-biased technical change. We conjecture that larger and/or FOR are more likely to import capital (technology) from abroad.

particular are paid more if employed in foreign-owned firms (quite possibly, they are barely paid any wage in local firms), and this may explain the unusual result for Ghana.

Table 5	n ownership	?			
	Cameroon	Ghana	Kenya	Zambia	Zimbabwe
Male	0.13	0.09	0.15	0.08	0.16
	(3.5) *	(1.8)	(4.4) *	(1.8)	(4.7) *
Age	0.08	0.18	0.04	0.05	0.10
	(4.8) *	(17.0) *	(3.6) *	(3.8) *	(9.2) *
Age-squared	-0.0007	-0.002	-0.0004	-0.0004	-0.001
	(-3.0) *	(-14.6) *	(-2.8) *	(-2.5) *	(-8.1) *
Tenure	0.011	0.01	0.007	0.03	-0.007
	(1.8)	(1.9)	(1.3)	(3.1) *	(-1.2)
Tenure-squared	-0.0001	-0.0003	-0.0000	-0.0009	0.0002
	(-0.4)	(-1.2)	(-0.0)	(-2.8) *	(1.4)
PRIMC	0.13	0.03	0.07	0.28	0.09
	(3.0) *	(0.4)	(2.4) *	(5.1) *	(2.4) *
SECC	0.49	0.18	0.20	0.67	0.43
	(10.3) *	(2.4) *	(5.5) *	(9.9) *	(8.0) *
UNIVC	1.24	0.88	1.19	1.62	0.86
	(17.5) *	(7.1) *	(9.8) *	(14.0) *	(4.3) *
Constant	8.1	5.6	8.1	8.1	3.3
	(27.1) *	(29.3) *	(38.4) *	(38.4) *	(16.0) *
State ownership	-0.03	0.01	-0.37	0.27	0.01
	(-0.5)	(0.2)	(-3.4) *	(4.3) *	(0.2)
Located in capital city	0.15	0.27	0.29	0.15	0.12
	(4.1) *	(7.5) *	(12.0) *	(4.1) *	(4.5) *
FOR * SKIL	0.18	0.03	0.34	0.33	0.07
	(3.6) *	(0.6)	(4.9) *	(4.8) *	(1.2)
FOR * UNSKILLED	0.03	0.28	0.16	0.11	0.11
	(0.8)	(7.6) *	(4.2) *	(1.9)	(2.4) *
Log(emp) * SKIL	0.16	0.23	0.15	0.15	0.22
	(10.3) *	(12.6) *	(11.5) *	(8.2) *	(17.6) *
Log(emp) * UNSKILLE	D 0.10	0.11	0.07	0.07	0.11
	(6.1) *	(6.7) *	(7.0) *	(4.4) *	(9.5) *
Time Dummies	Yes	Yes	Yes	Yes	Yes
Sector Dummies	Yes	Yes	Yes	Yes	Yes
Ν	1534	2257	1937	1593	1866
R-squared	0.55	0.52	0.45	0.50	0.48
Test H0: coefficients	P=0.016	P=0.000	P=0.027	P=0.0135	P=0.704
FOR*SK = FOR*UNSK					

able 5	Who	benefits	from	foreign	ownership?
	** 110	Denentito	nom	101 Cign	ownersmp.

Notes: As for Table 2. Dependent variable is log of monthly wages in current domestic currency.

Discussion

There is a large and significant wage premium associated with being employed in FOR. For all workers, without controls, this ranges from 20% in Cameron to 37% in Ghana and Zambia. The inclusion of firm specific controls reduces the premium, but it is still significant and ranges from 8% in Cameroon to about 22% in Ghana and Zambia. There is also an independent size wage-wage premium. There is thus convincing evidence that foreign-owned firms pay higher wages than local firms to workers with similar observable characteristics.

We can only speculate on why this might be so. Our results are in line with the technology explanation, such as provided by Strobl and Thornton (2001), if foreign firms are relatively technology intensive and skill-biased, at least during the early 90s. However, the technology explanation implies that foreign firms are more productive than local firms. The empirical evidence on this using the African data is mixed. Some researchers have found a significant and positive effect of the share of foreign ownership on technical efficiency in Ghana and Zimbabwe (Biggs *et al.*, 1995). Söderbom and Teal (2001b) did not find evidence for a significant effect of a foreign ownership dummy on technical efficiency, using system GMM estimators of production functions. Hence, the jury is still out on whether foreign firms are technically more efficient in specific cases and the results may vary by country, sector and perhaps by period of analysis. Furthermore, as our measure of foreign ownership refers to some ownership by a foreign national, and in few cases are FOR affiliates of multinationals, there is no presumption that they are more efficient.

To assess whether our results are consistent with a bargaining (rent-sharing) explanation we would need to demonstrate 1) that workers are better at capturing rents in foreignowned than in local firms and 2) that within foreign-owned firms skilled workers are better at bargaining than less-skilled workers. The bargaining explanation follows from the premise that workers in more profitable firms are paid higher wages because they can capture part of the rents (as formulated in Blanchflower *et al*, 1996). Söderbom and Teal (2001a) provide evidence in support of this in Ghana. We need to show that foreign firms are more profitable, to support point 1) above.

We use a specific database for Ghana, containing data on profits per employee and ownership, sector and location of the firms as well as the period (see Söderbom and Teal, 2001a, for a detailed description of this dataset). We regressed profits per employee (output minus material and labour costs per employee) on a foreign ownership dummy and controlled for other effects such as sector, location and period. For Ghana the average profit per employee is 36 percentage points higher in foreign firms than in local firms and the difference is significant (results available on request). If this were true for other African countries, and if skilled workers are indeed more effective at bargaining than less-skilled workers (because they are relatively scarce and more educated), our findings are consistent with a bargaining explanation. Further empirical work is required, however, to consolidate such an argument.

VI. CONCLUSIONS AND POLICY IMPLICATIONS

This paper used data on individual wages in the manufacturing industry of five African countries (Cameroon, Ghana, Kenya, Zambia and Zimbabwe) in the early 1990s to test whether foreign ownership is associated with higher wages for all education and occupation groups. As our measure of foreign is some ownership by a non-national (individual or multinational enterprise), we are in effect asking if wages differ in firms with access to foreign capital as compared to local firms (assumed to have more restricted access to capital). Consequently, we can assume that capital is cheaper for foreign-owned firms, therefore they will tend to use more capital intensive techniques and more skilled labour. We do not implicitly assume that foreign-owned firms are more efficient (on which the empirical evidence is mixed); while this may be expected for multinationals, it need not be the case for investment by foreign individuals.

We presented two main findings. First, we showed that foreign ownership is associated with a 20-37% differential in average wages (for all workers) in five African countries, allowing for the employee's age, education and tenure (i.e. observable workers characteristics). This FOR wage differential is halved to 8-23 per cent controlling for firm-specific effects (foreign-owned firms are larger and locate in high-wage sectors and

regions). Secondly, there is a tendency for skilled workers (as represented by occupation and education) to benefit more from foreign ownership than less skilled workers. Such skill-specific FOR differentials vary by country, and unskilled workers tend to receive higher wages if employed in FOR, although the benefit from having completed secondary education is apparent in all countries.

While foreign-owned firms pay higher wages than local firms to apparently equivalent employees, this tendency is strongest for more educated and skilled workers. Further empirical analysis is required to distinguish between the two explanations out forward in this paper, whether higher wages to skilled-workers in foreign-owned firms are due to their higher productivity or their greater bargaining power. A lower cost of capital explains why foreign-owned firms employ relatively more skilled workers (as they will use relatively capital and skill-intensive techniques), but does not in itself explain why they pay them more. It is not evident that foreign-owned firms are more efficient, nor is there any particular reason why this should be the case if ownership is by non-resident individuals. On the other hand, it is difficult to test explicitly if skilled workers in foreign firms are more effective in rent-seeking. As such workers are relatively scarce and education, there is a strong presumption that their bargaining power is greater.

Perhaps the two explanations are not mutually exclusive. Consider the dominant case of foreign-ownership, investment by a non-national. The employer will have a preference for relatively skilled labour, given the lower cost of capital, but may have less information about the local labour market. Skilled workers may know this. Thus, the employer is willing and able to pay higher wages, and the skilled employees gain a greater share of rents. Relative ignorance of the local labour market would imply that foreign owners also pay higher wages to unskilled workers and there is evidence of this. It is plausible, and consistent with the available evidence, to contend that foreign owners pay higher wages because they have less knowledge about local labour. Educated workers can obtain a higher premium from working in foreign firms because they have greater bargaining power. If workers *are* more productive in foreign-owned firms, this would compensate employers for paying higher wages but is not necessary to explain why they pay higher wages.

There is no evidence that any workers are paid less in foreign-owned firms, even unskilled workers benefit. While wages are higher in foreign-owned firms, we have no information on which to infer aggregate effects on incomes and/or employment, and hence on wage inequality. Whilst it seems likely that increased foreign ownership would tend to increase wage inequality, the presumption would be that average wages increase and employment is, at least, not reduced (foreign-owned firms tend to e larger). To contend that foreign investment (ownership) is harmful, one would need to demonstrate that it displaces local employment and has no positive spillover on wages in local firms. Restricting foreign ownership may well be counter-productive, as foreign firms tend to pay higher wages (*ceteris paribus*). On the other hand, encouraging foreign ownership (e.g. by attracting foreign investment, especially FDI) may not be an effective and efficient way to raise absolute and relative wages of low-income workers in the short-run. Complementary policies would be required to ensure that foreign ownership benefits all types of workers equally.

Based on the evidence in this paper, there appear to be at least two types of complementary policies. First, governments could bring the level of education of individuals up to completed first and secondary education. While this may reduce the wage premium as skilled workers are less scarce, it should increase employment as investors are attracted by the availability of skilled workers. Secondly, the government can support training schemes aimed at specific vocational training for less-skilled workers. While a general increase in skill levels may reduce the private return to education and training there should be a net social benefit through increased employment and, ideally, productivity.

However, any definite answer on the effects of foreign ownership on incomes of lowincome groups in these African countries would depend on information beyond what is available at present. For instance, based on the available data we were not able to specify what the effects were on the *level* of wages either in foreign or in local firms, but wage *differentials* between foreign and local firms only. Nor were we able to specify the direct employment effects, the indirect and long-run effects on wage and employment (especially in the informal sector), or the effects on non-manufacturing. Nevertheless, we are convinced by the evidence that employees of foreign-owned firms are paid higher wages than employees of locally-owned firms.

REFERENCES

- Aitken, B., A. Harrison and R. Lipsey (1996). "Wages and Foreign Ownership: A Comparative Study of Mexico, Venezuela, and the United States", *Journal of International Economics*, 42, pp. 345-371.
- Biggs, T., M. Shah and P. Srivatava (1995), "Technological Capabilities and Learning in African Enterprises", World Bank Technical Paper Number 2888, Africa Technical Department Series.
- Bigsten, A., P. Collier and others (1997), 'Exports of African Manufactures: Macro Policy and Firm Behaviour', Oxford: CSAE, mimeo.
- Bigsten, A., P. Collier and others (1999), 'Investment of Africa's Manufacturing Sector: a Four Country Panel Data Analysis', Oxford Bulletin of Economics and Statistics, 61, pp. 489-512.
- Bigsten, A., A. Isakson and others (2000), "Rates of Return on Physical and Human Capital in Africa's Manufacturing Sector", *Economic Development and Cultural Change*, 48, pp. 801-827.
- Blanchflower, D.G., A.J. Oswald and P. Sanfey (1996), "Wages, profits and rent-sharing", *Quarterly Journal of Economics*, 111, pp. 227-251.
- Blundell, R. and A. Duncan (1998), "Kernel regression in empirical microeconomics", *Journal of Human Resources*, winter 1998, 33, pp. 62-87.
- BusinessMap (2000). SADC Investor Survey: Complex Terrain, BusinessMap, Johannesburg.
- Dunning, J. (1993). *Multinational enterprises and the global economy*, Addison-Wesley Publishing Company.
- Feenstra, R.C and G.H. Hanson (1995), "Foreign Direct Investment and Relative Wages: Evidence from Mexico's Maquiladoras", *NBER working paper 5122*.
- Fujita, M., P. Krugman and A.J. Venables (2000), "*The Spatial Economy. Cities, Regions and International Trade*, MIT Press, Cambridge, Massachusetts.
- Görg, H. and E. Strobl (2001), 'Relative Wages, Openness and Skill-Biased Technological Change in Ghana', University of Nottingham, School of Economics, CREDIT Research Paper (forthcoming).
- Lipsey, B. and F. Sjöholm (2001), "Foreign Direct Investment and Wages in Indonesian Manufacturing, *NBER working paper 8299*.
- Mazumdar, D. (1995), "Wage differences by size of enterprise in African manufacturing", *RPED Discussion Paper*, August 1995, World Bank.
- Mincer, J. (1974), Schooling, Experience, and Earnings, NBER, New York.
- Polachek, S.W. and W.S. Siebert (1993), *The Economics of Earnings*, Cambridge University Press, Cambridge.
- Söderbom, M. and F. Teal (2001a), "Firm size and human capital as determinants of productivity and earnings", Oxford University: *CSAE Working Paper WPS/2001-09*

- Söderbom, M. and F. Teal (2001b), "Are African Manufacturing Firms Really Inefficient? Evidence from Firm-Level Panel Data", Oxford University: CSAE Working Paper WPS/2001-14
- Strobl, E. and B. Thornton (2001), "Do Large Employers Pay More? A Study of Five African Countries", paper presented at the DESG Conference 2001, University of Nottingham, 5-7 April.
- Teal, F. (2000), "Private Sector Wages and Poverty in Ghana: 1988-1998", Oxford University: CSAE Working Paper WPS/2000-6.
- UNCTAD (2000), World Investment Report 2000, UNCTAD, Geneva.
- Velde, te D.W. (2001), "Policies towards Foreign Direct Investment in Developing Countries: Emerging Best-Practices and Outstanding Issues", mimeo, ODI, March 2001, downloadable from http://www.odi.org.uk/iedg/FDI_Conference/FDIhome.html
- White, H. (1980), "A heteroskedasticity-consistent covariance matrix estimator and a direct test for heteroskedasticity", *Econometrica*, 48, pp. 817-838.

APPENDIX A Mincer's Human Capital Earnings Functions

The human capital earnings function usually referred to as the standard human capital earnings function of Mincer (1974) is:

(A1)
$$\ln Y_{it} = \alpha + rS_i + \beta_1 t_i + \beta_2 t_i^2 + u_{it}$$

where $(\ln Y)$ is the log of earnings for individual *i* at time *t*, S_i is number of years spent at school and t_i reflects post-school experience. The estimate of *r* can be seen as the rate of return to one more year of schooling.

The derivation of Mincer's equation A1 was originally based on human capital theory that a worker maximises discounted lifetime earnings, and that there is a choice between time spent on *current* earnings and time spent on learning (an investment that generates *future* earnings). The positive relationship between S and Y arises because schooling raises a worker's productivity and hence wages. This can be contrasted with a signalling or sorting explanation which explains a positive relationship between S and Y by the fact that education is correlated with real productivity-enhancing attributes (ability etc.), hence a signal on which to sort individuals. Equation A1 can also be seen as a simple accounting identity (and not based on human capital theory) in that earnings at t+1 are earnings at t plus the investment made in human capital times the rate of return on the investment.

There are a number of potential biases when estimating A1 by OLS, although these have not been shown to be serious in practice in the African case. First, S may be endogenous. In an optimising investment model we expect that schooling is positively correlated with the return to schooling and hence the OLS estimate of r is biased upwards. Secondly, an OLS estimate of r is biased because the schooling is often measured with error. Thirdly, an OLS estimate of r is biased upwards due to unobserved worker attributes with are correlated positively with schooling and earnings, often called the ability bias. Finally, the effect of schooling may vary systematically with family background characteristics that also affect the number of years of schooling. We acknowledge these problems. However, we do not specifically interpret the coefficient r, but focus on the effects of foreign ownership. As such it may be that foreign ownership captures ability or other bias, for instance, if more able individuals are employed by foreign-owned firms.
Country	V	Vave 1	W	Vave 2	Wave 3		
	LOC	FOR	LOC	FOR	LOC	FOR	
Cameroon							
Wages	383.18	629.82	427.32	695.70	295.92	476.48	
N (workers)	579	403	450	273	335	200	
Primary	.5095	.5079	.4089	.4135	.3413	.2037	
Secondary	.3251	.3395	.4427	.4346	.4960	.6420	
University	.0628	.0526	.0703	.0802	.0992	.1111	
Tenure	5.2199	8.7879	4.5761	9.7838	5.3204	9.6060	
Age	33.0399	36.0605	31.3438	36.6076	33.1468	36.512	
Firm size	3.1116	4.6409	2.9693	4.4362	2.8620	4.2797	
Ghana							
Wages	153.868	231.983	155.098	256.125	135.008	247.693	
N (workers)	541	143	574	185	889	225	
Primary	.7375	.6763	.7651	.7717	.7037	.7265	
Secondary	.1950	.2374	.1706	.1957	.2025	.2242	
University	.0077	.0360	.0147	.0217	.0069	.0224	
Tenure	5.107	8.034	5.160	8.190	4.441	7.781	
Age	29.736	35.734	28.842	35.082	28.856	36.628	
Firm size	3.212	4.365	3.198	4.345	3.330	4.548	
Kenya							
Wages	297.366	393.885	230.547	484.285	350.517	645.294	
N (workers)	886	299	833	212	832	200	
Primary	.469	.332	.426	.316	.470	.345	
Secondary	.343	.430	.351	.435	.371	.452	
University	.007	.013	.007	.043	.021	.076	
Tenure	7.531	9.432	6.830	9.184	7.268	8.614	
Age	33.808	36.198	32.310	35.474	32.859	34.751	
Firm size	3.684	4.615	3.568	4.608	3.609	4.648	

Table A1 Summary Statistics: Principal variables by Country

2	\mathbf{a}
- 4	,
2	4

Country	 W	Wave 1			Wave 2 Wave 3		
Country			LOC				
 Zambia							
Wages	175.550	281.037	190.882	260.121	141.729	240.794	
N (workers)	641	171	537	117	273	92	
Primary	.5305	.3567	.5194	.3652	.4688	.4719	
Secondary	.3376	.5146	.3527	.4261	.4297	.4719	
University	.0370	.0702	.0426	.1043	.0391	.0449	
Tenure	5.979	8.128	6.414	9.001	6.973	8.699	
Age	33.330	36.298	33.295	37.452	34.293	37.191	
Firm size	3.705	5.174	3.589	4.141	3.902	4.753	
Zimbabwe							
Wages	290.905	474.258	273.967	506.180			
N (workers)	1,012	374	452	173			
Primary	.4745	.4595	.5309	.4452			
Secondary	.3457	.3216	.3041	.2710			
University	.010	.035	.008	.045			
Tenure	8.832	12.277	9.350	12.443			
Age	34.329	37.549	35.119	38.239			
Firm size							

Variables are mean of monthly wages in US\$ PPP, number of workers (N), percentage with primary education completed, secondary education completed and university education, mean tenure within firm (years), mean age of individuals (years) and log of firm employment. FOR = foreign-owned firm; LOC = local firm.

	(1)	(2)	(3)	(4)	(5)
Male	0.02	0.01	0.03	0.03	0.04
	(0.6)	(0.4)	(0.8)	(0.7)	(1.0)
Age	0.12	0.11	0.08	0.08	0.08
	(6.3)*	(5.8) *	(4.5) *	(4.5) *	(4.7) *
Age-squared	-0.001	-0.001	-0.0006	-0.0006	-0.0007
	(-4.5)*	(-4.0) *	(-2.7) *	(-2.7) *	(-2.9) *
Tenure	0.03	0.02	0.01	0.01	0.01
	(5.1) *	(3.6) *	(1.7)	(1.7)	(1.7)
Tenure-squared	-0.0004	-0.0002	-0.0001	-0.0001	-0.0001
	(-1.6)	(-0.8)	(-0.5)	(-0.6)	(-0.5)
Primary education	0.20	0.16	0.16	0.15	0.16
completed (PRIMC)	(4.1) *	(3.4)*	(3.4) *	(3.4) *	(2.9) *
Secondary education	0.72	0.70	0.59	0.59	0.53
completed (SECC)	(14.1) *	(14.0) *	(12.0) *	(12.0) *	(9.1) *
University completed	1.57	1.53	1.36	1.36	1.31
(UNIVC)	(22.2) *	(21.7) *	(19.2) *	(19.1) *	(15.5) *
Constant	7.8	8.0	8.1	8.1	8.1
	(22.9) *	(23.0) *	(26.3) *	(26.3) *	(26.3) *
Log (employment)			0.13	0.13	0.13
			(8.8) *	(8.7) *	(8.8) *
Foreign ownership		0.20	0.08		
(FOF)		(6.6) *	(2.4) *		
State ownership			0.01	-0.05	0.02
			(0.2)	(-0.1)	(0.3)
Located in capital city			0.15	0.15	0.16
			(4.0) *	(4.1) *	(4.1) *
Time dummies	Yes	Yes	Yes	Yes	Yes
Sector Dummies			Yes	Yes	Yes
FOR*sector Dummies				Yes	
FOR*Educ Dummies					Yes
N	1655	1631	1534	1534	1534
R-squared	0.47	0.48	0.52	0.52	0.52
Test				P=0.52	P=0.07

Table A2Wage equations for Cameroon

Notes: Dependent variable is log of monthly wages in current domestic currency. Equations as specified in the text. White heteroscedasticity-consistent *t*-statistics in parenthesis; * indicates significance at least at the 5% level.

	(1)	(2)	(3)	(4)	(5)
Male	-0.02	-	0.03	0.03	0.03
	(-0.5)	0.03	(0.5)	(0.5)	(0.5)
		(-			
		0.1)			
Age	0.22	0.22	0.19	0.22	0.19
	(22.4) *	(21.	(17.7) *	(22.4) *	(17.5) *
		0) *			
Age-squared	-0.002	-0.002	-0.002	-0.002	-0.002
	(-18.5) *	(-17.6) *	(-14.9) *	(-14.8) *	(-14.7) *
Tenure	0.02	0.01	0.01	0.01	0.01
	(2.5) *	(1.7)	(1.8)	(1.9)	(1.8)
Tenure-squared	-0.0003	-0.0001	-0.0001	-0.0001	-0.0001
	(-1.1)	(-0.6)	(-0.6)	(-0.8)	(-0.6)
Primary education	0.25	0.17	0.08	0.08	0.06
completed (PRIMC)	(3.9) *	(2.5) *	(1.1)	(1.2)	(0.9)
Secondary education	0.57	0.47	0.32	0.32	0.31
completed (SECC)	(8.2) *	(6.4) *	(4.3) *	(4.4) *	(3.9) *
University completed	1.40	1.39	1.18	1.19	1.44
(UNIVC)	(12.7) *	(10.7) *	(9.4) *	(9.3) *	(10.7) *
Constant	5.1	5.2	5.3	5.3	5.3
	(27.1) *	(26.3) *	(27.5) *	(27.3) *	(27.2) *
Log (employment)			0.15	0.16	0.15
			(9.1) *	(9.0) *	(9.1) *
Foreign ownership		0.37	0.22		
(FOF)		(12.4) *	(6.6) *		
State ownership			0.03	0.03	0.03
			(0.7)	(0.6)	(0.7)
Located in capital			0.26	0.27	0.26
city			(7.2) *	(7.2) *	(7.2) *
Time Dummies	Yes	Yes	Yes	Yes	Yes
Sector Dummies			Yes	Yes	Yes
FOR*Sector				Yes	
Dummies					
FOR*Educ Dummies					Yes
N	2557	2257	2257	2257	2257
R-squared	0.45	0.47	0.50	0.51	0.51
Test				P=0.43	P=0.00

Table A3Wage Equations for Ghana

	(1)	(2)	(3)	(4)	(5)
Male	0.06	0.05	0.08	0.08	0.08
	(1.9)	(1.6)	(2.6) *	(2.7) *	(2.4) *
Age	0.06	0.05	0.04	0.04	0.04
-	(5.4) *	(5.4) *	(3.5) *	(3.5) *	(3.5) *
Age-squared	-0.0005	-0.0005	-0.0003	-0.0003	-0.0003
	(-3.5) *	(-3.0) *	(-2.0) *	(-2.0) *	(-1.9)
Tenure	0.04	0.006	0.0001	0.0001	0.0002
	(0.8)	(1.1)	(0.0)	(0.0)	(0.0)
Tenure-squared	0.0001	-0.0001	0.0001	0.0001	0.0001
	(0.3)	(-0.3)	(0.5)	(0.6)	(0.5)
Primary education	0.21	0.21	0.16	0.16	0.14
completed (PRIMC)	(7.6) *	(7.5) *	(6.1) *	(6.0) *	(4.8) *
Secondary education	0.58	0.55	0.43	0.42	0.35
completed (SECC)	(18.3) *	(17.1) *	(13.8) *	(13.7) *	(10.6) *
University completed	1.71	1.61	1.42	1.42	1.28
(UNIVC)	(15.8) *	(15.2) *	(13.0) *	(13.4) *	(10.3) *
Constant	6.2	6.2	6.1	6.1	6.2
	(32.4) *	(32.2) *	(34.2) *	(33.9) *	(34.7) *
Log (employment)			0.09	0.09	0.09
			(10.4) *	(10.4) *	(10.7) *
Foreign ownership		0.24	0.17		
(FOR)		(8.7) *	(6.0) *		
State ownership			-0.07	0.06	-0.12
			(-0.6)	(0.5)	(-1.1)
Located in capital city			0.28	0.27	0.28
			(13.8) *	(12.2) *	(13.5) *
Time Dummies	Yes	Yes	Yes	Yes	Yes
Sector Dummies			Yes	Yes	Yes
FOR*Sector Dummies				Yes	
FOR*Educ Dummies					Yes
N	3143	3045	3035	3035	3035
R-squared	0.29	0.29	0.38	0.39	0.39
Test				P=0.00	P=0.00

Table A4Wage Equations for Kenya

	(1)	(2)	(3)	(4)	(5)
Male	0.01	-0.00	-0.01	-0.02	-0.01
Maic	(0.3)	(-0.0)	(-0.2)	(-0.0)	(-0.2)
A a a	0.08	0.06	0.05	0.05	0.05
Age	(6.0) *	(6.0) *	(4.3) *	(4.0) *	(4.3) *
A					
Age-squared	-0.0008	-0.0006	-0.0005	-0.0005	-0.0005 (-4.5) *
Τ	(-4.5) *	(-3.6) *	(-2.8) *	(-2.6) *	. ,
Tenure	0.04	0.04	0.03	0.03	0.03
T	(5.2) *	(4.1) *	(3.6) *	(3.5) *	(3.7) *
Tenure-squared	-0.001	-0.001	-0.001	-0.001	-0.001
	(-3.9) *	(-3.3) *	(-2.9) *	(-2.8) *	(-2.9) *
Primary education	0.36	0.36	0.34	0.34	0.37
completed (PRIMC)	(5.6) *	(6.3) *	(5.9) *	(5.7) *	(5.9) *
Secondary education	1.01	0.99	0.84	0.84	0.81
completed (SECC)	(14.4) *	(15.4) *	(12.3) *	(12.1) *	(11.2) *
University completed	2.13	2.06	1.88	1.87	1.82
(UNIVC)	(20.8) *	(19.9) *	(16.3) *	(16.0) *	(20.8) *
Constant	8.0	8.2	7.9	7.9	7.9
	(37.3) *	(39.4) *	(36.4) *	(36.1) *	(35.9) *
Log (employment)			0.10	0.11	0.11
			(5.9) *	(6.3) *	(6.1) *
Foreign ownership		0.37	0.23		
(FOR)		(8.4) *	(4.7) *		
State ownership			0.24	0.24	0.25
			(3.8) *	(3.6) *	(3.9) *
Located in capital city			0.15	0.17	0.16
			(4.0) *	(4.4) *	(4.2) *
Time Dummies	Yes	Yes	Yes	Yes	Yes
Sector Dummies			Yes	Yes	Yes
FOR*Sector Dummies				Yes	
FOF*Educ Dummies					Yes
N	2471	1763	1593	1593	1593
R-squared	0.39	0.44	0.47	0.48	0.48
Test				P=0.02	P=0.16

Table A5Wage Equations for Zambia

	(1)	(2)	(3)	(4)	(5)
Male	0.21	0.18	0.11	0.10	0.12
	(5.1) *	(4.3) *	(3.0) *	(2.6) *	(3.0) *
Age	0.16	0.15	0.13	0.13	0.13
	(12.9) *	(12.6) *	(11.0) *	(10.9) *	(10.9) *
Age-squared	-0.002	-0.002	-0.001	-0.001	-0.001
	(-11.6) *	(-11.3) *	(-9.8) *	(-9.6) *	(-9.6) *
Tenure	0.01	0.03	-0.01	-0.01	-0.01
	(2.3) *	(0.5)	(-1.8)	(-2.0) *	(-1.5)
Tenure-squared	-0.0004	0.0002	0.0005	0.0005	0.0004
	(-0.2)	(1.0)	(2.6) *	(2.6) *	(2.4) *
Primary education	0.15	0.15	0.14	0.16	0.12
completed (PRIMC)	(3.5) *	(3.4) *	(3.5) *	(3.9) *	(2.5) *
Secondary education	0.77	0.73	0.58	0.58	0.53
completed (SECC)	(13.0) *	(12.1) *	(10.2) *	(10.3) *	(8.4) *
University completed	1.79	1.64	1.27	1.28	1.73
(UNIVC)	(10.5) *	(9.3) *	(6.8) *	(7.2) *	(20.5) *
Constant	2.6	2.7	2.6	2.6	2.7
	(11.3) *	(11.8) *	(11.7) *	(11.8) *	(12.1) *
Log (employment)			0.14	0.15	0.14
			(12.2) *	(8.7) *	(11.9) *
Foreign ownership		0.30	0.13		
(FOR)		(8.3) *	(3.3) *		
State ownership			0.06	0.05	0.05
			(0.8)	(0.7)	(0.6)
Located in capital city			0.12	0.08	0.12
			(4.1) *	(2.8) *	(4.0)
TimeDummies	Yes	Yes	Yes	Yes	Yes
Sector Dummies			Yes	Yes	Yes
FOR*Sector Dummies				Yes	
FOR*Educ Dummies					Yes
N	1960	1866	1866	1866	1866
R-squared	0.29	0.30	0.38	0.39	0.39
Test				P=0.00	P=0.004

Table A6Wage Equations for Zimbabwe

CREDIT PAPERS

- 99/1 **Ewen Cummins,** "Hey and Orme go to Gara Godo: Household Risk Preferences"
- 99/2 Louise Grenier, Andrew McKay and Oliver Morrissey, "Competition and Business Confidence in Manufacturing Enterprises in Tanzania"
- 99/3 **Robert Lensink and Oliver Morrissey,** "Uncertainty of Aid Inflows and the Aid-Growth Relationship"
- 99/4 **Michael Bleaney and David Fielding,** "Exchange Rate Regimes, Inflation and Output Volatility in Developing Countries"
- 99/5 **Indraneel Dasgupta**, "Women's Employment, Intra-Household Bargaining and Distribution: A Two-Sector Analysis"
- 99/6 Robert Lensink and Howard White, "Is there an Aid Laffer Curve?"
- 99/7 **David Fielding,** "Income Inequality and Economic Development: A Structural Model"
- 99/8 Christophe Muller, "The Spatial Association of Price Indices and Living Standards"
- 99/9 **Christophe Muller,** "The Measurement of Poverty with Geographical and Intertemporal Price Dispersion"
- 99/10 Henrik Hansen and Finn Tarp, "Aid Effectiveness Disputed"
- 99/11 Christophe Muller, "Censored Quantile Regressions of Poverty in Rwanda"
- 99/12 Michael Bleaney, Paul Mizen and Lesedi Senatla, "Portfolio Capital Flows to Emerging Markets"
- 99/13 **Christophe Muller**, "The Relative Prevalence of Diseases in a Population of Ill Persons"
- 00/1 **Robert Lensink,** "Does Financial Development Mitigate Negative Effects of Policy Uncertainty on Economic Growth?"
- 00/2 **Oliver Morrissey,** "Investment and Competition Policy in Developing Countries: Implications of and for the WTO"
- 00/3 Jo-Ann Crawford and Sam Laird, "Regional Trade Agreements and the WTO"
- 00/4 **Sam Laird,** "Multilateral Market Access Negotiations in Goods and Services"
- 00/5 Sam Laird, "The WTO Agenda and the Developing Countries"
- 00/6 Josaphat P. Kweka and Oliver Morrissey, "Government Spending and Economic Growth in Tanzania, 1965-1996"
- 00/7 Henrik Hansen and Fin Tarp, "Aid and Growth Regressions"
- 00/8 Andrew McKay, Chris Milner and Oliver Morrissey, "The Trade and Welfare Effects of a Regional Economic Partnership Agreement"
- 00/9 Mark McGillivray and Oliver Morrissey, "Aid Illusion and Public Sector Fiscal Behaviour"
- 00/10 **C.W. Morgan,** "Commodity Futures Markets in LDCs: A Review and Prospects"
- 00/11 Michael Bleaney and Akira Nishiyama, "Explaining Growth: A Contest between Models"

- 00/12 **Christophe Muller**, "Do Agricultural Outputs of Autarkic Peasants Affect Their Health and Nutrition? Evidence from Rwanda"
- 00/13 **Paula K. Lorgelly,** "Are There Gender-Separate Human Capital Effects on Growth? A Review of the Recent Empirical Literature"
- 00/14 **Stephen Knowles and Arlene Garces,** "Measuring Government Intervention and Estimating its Effect on Output: With Reference to the High Performing Asian Economies"
- 00/15 I. Dasgupta, R. Palmer-Jones and A. Parikh, "Between Cultures and Markets: An Eclectic Analysis of Juvenile Gender Ratios in India"
- 00/16 **Sam Laird,** "Dolphins, Turtles, Mad Cows and Butterflies A Look at the Multilateral Trading System in the 21st Century"
- 00/17 Carl-Johan Dalgaard and Henrik Hansen, "On Aid, Growth, and Good Policies"
- 01/01 **Tim Lloyd, Oliver Morrissey and Robert Osei,** "Aid, Exports and Growth in Ghana"
- 01/02 **Christophe Muller,** "Relative Poverty from the Perspective of Social Class: Evidence from The Netherlands"
- 01/03 **Stephen Knowles,** "Inequality and Economic Growth: The Empirical Relationship Reconsidered in the Light of Comparable Data"
- 01/04 **A. Cuadros, V. Orts and M.T. Alguacil,** "Openness and Growth: Re-Examining Foreign Direct Investment and Output Linkages in Latin America"
- 01/05 Harold Alderman, Simon Appleton, Lawrence Haddad, Lina Song and Yisehac Yohannes, "Reducing Child Malnutrition: How Far Does Income Growth Take Us?"
- 01/06 **Robert Lensink and Oliver Morrissey,** "Foreign Direct Investment: Flows, Volatility and Growth"
- 01/07 Adam Blake, Andrew McKay and Oliver Morrissey, "The Impact on Uganda of Agricultural Trade Liberalisation"
- 01/08 **R. Quentin Grafton, Stephen Knowles and P. Dorian Owen,** "Social Divergence and Economic Performance"
- 01/09 **David Byrne and Eric Strobl,** "Defining Unemployment in Developing Countries: The Case of Trinidad and Tobago"
- 01/10 **Holger Görg and Eric Strobl,** "The Incidence of Visible Underemployment: Evidence for Trinidad and Tobago"
- 01/11 **Abbi Mamo Kedir,** "Some Issues in Using Unit Values as Prices in the Estimation of Own-Price Elasticities: Evidence from Urban Ethiopia"
- 01/12 Eric Strobl and Frank Walsh, "Minimum Wages and Compliance: The Case of Trinidad and Tobago"
- 01/13 Mark McGillivray and Oliver Morrissey, "A Review of Evidence on the Fiscal Effects of Aid"
- 01/14 **Tim Lloyd, Oliver Morrissey and Robert Osei,** "Problems with Pooling in Panel Data Analysis for Developing Countries: The Case of Aid and Trade Relationships"
- 01/15 **Oliver Morrissey,** "Pro-Poor Conditionality for Aid and Debt Relief in East Africa"

- 01/16 Zdenek Drabek and Sam Laird, "Can Trade Policy help Mobilize Financial Resources for Economic Development?"
- 01/17 **Michael Bleaney and Lisenda Lisenda,** "Monetary Policy After Financial Liberalisation: A Central Bank Reaction Function for Botswana"
- 01/18 **Holger Görg and Eric Strobl,** "Relative Wages, Openness and Skill-Biased Technological Change in Ghana"
- 01/19 **Dirk Willem te Velde and Oliver Morrissey,** "Foreign Ownership and Wages: Evidence from Five African Countries"

DEPARTMENT OF ECONOMICS DISCUSSION PAPERS

In addition to the CREDIT series of research papers the School of Economics produces a discussion paper series dealing with more general aspects of economics. Below is a list of recent titles published in this series.

- 99/1 Indraneel Dasgupta, "Stochastic Production and the Law of Supply"
- 99/2 Walter Bossert, "Intersection Quasi-Orderings: An Alternative Proof"
- 99/3 Charles Blackorby, Walter Bossert and David Donaldson, "Rationalizable Variable-Population Choice Functions"
- 99/4 **Charles Blackorby, Walter Bossert and David Donaldson**, "Functional Equations and Population Ethics"
- 99/5 **Christophe Muller**, "A Global Concavity Condition for Decisions with Several Constraints"
- 99/6 **Christophe Muller**, "A Separability Condition for the Decentralisation of Complex Behavioural Models"
- 99/7 **Zhihao Yu**, "Environmental Protection and Free Trade: Indirect Competition for Political Influence"
- 99/8 Zhihao Yu, "A Model of Substitution of Non-Tariff Barriers for Tariffs"
- 99/9 **Steven J. Humphrey,** "Testing a Prescription for the Reduction of Non-Transitive Choices"
- 99/10 **Richard Disney, Andrew Henley and Gary Stears,** "Housing Costs, House Price Shocks and Savings Behaviour Among Older Households in Britain"
- 99/11 Yongsheng Xu, "Non-Discrimination and the Pareto Principle"
- 99/12 Yongsheng Xu, "On Ranking Linear Budget Sets in Terms of Freedom of Choice"
- 99/13 Michael Bleaney, Stephen J. Leybourne and Paul Mizen, "Mean Reversion of Real Exchange Rates in High-Inflation Countries"
- 99/14 Chris Milner, Paul Mizen and Eric Pentecost, "A Cross-Country Panel Analysis of Currency Substitution and Trade"
- 99/15 Steven J. Humphrey, "Are Event-splitting Effects Actually Boundary Effects?"
- 99/16 Taradas Bandyopadhyay, Indraneel Dasgupta and Prasanta K. Pattanaik, "On the Equivalence of Some Properties of Stochastic Demand Functions"
- 99/17 Indraneel Dasgupta, Subodh Kumar and Prasanta K. Pattanaik, "Consistent Choice and Falsifiability of the Maximization Hypothesis"
- 99/18 **David Fielding and Paul Mizen,** "Relative Price Variability and Inflation in Europe"
- 99/19 Emmanuel Petrakis and Joanna Poyago-Theotoky, "Technology Policy in an Oligopoly with Spillovers and Pollution"
- 99/20 **Indraneel Dasgupta,** "Wage Subsidy, Cash Transfer and Individual Welfare in a Cournot Model of the Household"
- 99/21 Walter Bossert and Hans Peters, "Efficient Solutions to Bargaining Problems with Uncertain Disagreement Points"
- 99/22 Yongsheng Xu, "Measuring the Standard of Living An Axiomatic Approach"

- 99/23 Yongsheng Xu, "No-Envy and Equality of Economic Opportunity"
- 99/24 M. Conyon, S. Girma, S. Thompson and P. Wright, "The Impact of Mergers and Acquisitions on Profits and Employee Remuneration in the United Kingdom"
- 99/25 **Robert Breunig and Indraneel Dasgupta**, "Towards an Explanation of the Cash-Out Puzzle in the US Food Stamps Program"
- 99/26 John Creedy and Norman Gemmell, "The Built-In Flexibility of Consumption Taxes"
- 99/27 **Richard Disney,** "Declining Public Pensions in an Era of Demographic Ageing: Will Private Provision Fill the Gap?"
- 99/28 Indraneel Dasgupta, "Welfare Analysis in a Cournot Game with a Public Good"
- 99/29 Taradas Bandyopadhyay, Indraneel Dasgupta and Prasanta K. Pattanaik, "A Stochastic Generalization of the Revealed Preference Approach to the Theory of Consumers' Behavior"
- 99/30 Charles Blackorby, WalterBossert and David Donaldson, "Utilitarianism and the Theory of Justice"
- 99/31 **Mariam Camarero and Javier Ordóñez,** "Who is Ruling Europe? Empirical Evidence on the German Dominance Hypothesis"
- 99/32 Christophe Muller, "The Watts' Poverty Index with Explicit Price Variability"
- 99/33 Paul Newbold, Tony Rayner, Christine Ennew and Emanuela Marrocu, "Testing Seasonality and Efficiency in Commodity Futures Markets"
- 99/34 **Paul Newbold, Tony Rayner, Christine Ennew and Emanuela Marrocu,** "Futures Markets Efficiency: Evidence from Unevenly Spaced Contracts"
- 99/35 **Ciaran O'Neill and Zoe Phillips,** "An Application of the Hedonic Pricing Technique to Cigarettes in the United Kingdom"
- 99/36 Christophe Muller, "The Properties of the Watts' Poverty Index Under Lognormality"
- 99/37 **Tae-Hwan Kim, Stephen J. Leybourne and Paul Newbold,** "Spurious Rejections by Perron Tests in the Presence of a Misplaced or Second Break Under the Null"
- 00/1 Tae-Hwan Kim and Christophe Muller, "Two-Stage Quantile Regression"
- 00/2 **Spiros Bougheas, Panicos O. Demetrides and Edgar L.W. Morgenroth,** "International Aspects of Public Infrastructure Investment"
- 00/3 Michael Bleaney, "Inflation as Taxation: Theory and Evidence"
- 00/4 Michael Bleaney, "Financial Fragility and Currency Crises"
- 00/5 **Sourafel Girma,** "A Quasi-Differencing Approach to Dynamic Modelling from a Time Series of Independent Cross Sections"
- 00/6 **Spiros Bougheas and Paul Downward,** "The Economics of Professional Sports Leagues: A Bargaining Approach"
- 00/7 Marta Aloi, Hans Jørgen Jacobsen and Teresa Lloyd-Braga, "Endogenous Business Cycles and Stabilization Policies"
- 00/8 **A. Ghoshray, T.A. Lloyd and A.J. Rayner,** "EU Wheat Prices and its Relation with Other Major Wheat Export Prices"

- 00/9 **Christophe Muller**, "Transient-Seasonal and Chronic Poverty of Peasants: Evidence from Rwanda"
- 00/10 **Gwendolyn C. Morrison,** "Embedding and Substitution in Willingness to Pay"
- 00/11 **Claudio Zoli**, "Inverse Sequential Stochastic Dominance: Rank-Dependent Welfare, Deprivation and Poverty Measurement"
- 00/12 **Tae-Hwan Kim, Stephen Leybourne and Paul Newbold,** "Unit Root Tests With a Break in Variance"
- 00/13 **Tae-Hwan Kim, Stephen Leybourne and Paul Newbold**, "Asymptotic Mean Squared Forecast Error When an Autoregression With Linear Trend is Fitted to Data Generated by an I(0) or I(1) Process"
- 00/14 **Michelle Haynes and Steve Thompson,** "The Productivity Impact of IT Deployment: An Empirical Evaluation of ATM Introduction"
- 00/15 Michelle Haynes, Steve Thompson and Mike Wright, "The Determinants of Corporate Divestment in the UK"
- 00/16 John Beath, Robert Owen, Joanna Poyago-Theotoky and David Ulph, "Optimal Incentives for Incoming Generations within Universities"
- 00/17 S. McCorriston, C. W. Morgan and A. J. Rayner, "Price Transmission: The Interaction Between Firm Behaviour and Returns to Scale"
- 00/18 **Tae-Hwan Kim, Douglas Stone and Halbert White,** "Asymptotic and Bayesian Confidence Intervals for Sharpe Style Weights"
- 00/19 **Tae-Hwan Kim and Halbert White**, "James-Stein Type Estimators in Large Samples with Application to the Least Absolute Deviation Estimator"
- 00/20 **Gwendolyn C. Morrison,** "Expected Utility and the Endowment Effect: Some Experimental Results"
- 00/21 **Christophe Muller**, "Price Index Distribution and Utilitarian Social Evaluation Functions"
- 00/22 **Michael Bleaney,** "Investor Sentiment, Discounts and Returns on Closed-End Funds"
- 00/23 **Richard Cornes and Roger Hartley,** "Joint Production Games and Share Functions"
- 00/24 **Joanna Poyago-Theotoky,** "Voluntary Approaches, Emission Taxation and the Organization of Environmental R&D"
- 00/25 Michael Bleaney, Norman Gemmell and Richard Kneller, "Testing the Endogenous Growth Model: Public Expenditure, Taxation and Growth Over the Long-Run"
- 00/26 **Michael Bleaney and Marco Gundermann,** "Credibility Gains and Output Losses: A Model of Exchange Rate Anchors"
- 00/27 **Indraneel Dasgupta**, "Gender Biased Redistribution and Intra-Household Distribution"
- 00/28 **Richard Cornes and Roger Hartley,** "Rentseeking by Players with Constant Absolute Risk Aversion"
- 00/29 S.J. Leybourne, P. Newbold, D. Vougas and T. Kim, "A Direct Test for Cointegration Between a Pair of Time Series"
- 00/30 **Claudio Zoli,** "Inverse Stochastic Dominance, Inequality Measurement and Gini Indices"

- 01/01 Spiros Bougheas, "Optimism, Education, and Industrial Development"
- 01/02 **Tae-Hwan Kim and Paul Newbold**, "Unit Root Tests Based on Inequality-Restricted Estimators"
- 01/03 Christophe Muller, "Defining Poverty Lines as a Fraction of Central Tendency"
- 01/04 **Claudio Piga and Joanna Poyago-Theotoky**, "Shall We Meet Halfway? Endogenous Spillovers and Locational Choice"
- 01/05 **Ilias Skamnelos,** "Sunspot Panics, Information-Based Bank Runs and Suspension of Deposit Convertibility"
- 01/06 **Spiros Bougheas and Yannis Georgellis,** "Apprenticeship Training, Earnings Profiles and Labour Turnover: Theory and German Evidence"
- 01/07 **M.J. Andrews, S. Bradley and R. Upward,** "Employer Search, Vacancy Duration and Skill Shortages"
- 01/08 Marta Aloi and Laurence Lasselle, "Growing Through Subsidies"
- 01/09 **Marta Aloi and Huw D. Dixon,** "Entry Dynamics, Capacity Utilisation, and Productivity in a Dynamic Open Economy"
- 01/10 **Richard Cornes and Roger Hartley,** "Asymmetric Contests with General Technologies"
- 01/11 Richard Cornes and Roger Hartley, "Disguised Aggregative Games"

Members of the Centre

Director

Oliver Morrissey - aid policy, trade and agriculture

Research Fellows (Internal)

Simon Appleton – poverty, education, households Adam Blake – CGE models of low-income countries Mike Bleaney - growth, international macroeconomics Indraneel Dasgupta – development theory Norman Gemmell – growth and public sector issues Ken Ingersent - agricultural trade Tim Lloyd – agricultural commodity markets Paula Lorgelly – health, gender and growth Andrew McKay - poverty, peasant households, agriculture Chris Milner - trade and development Wyn Morgan - futures markets, commodity markets Christophe Muller – poverty, household panel econometrics Tony Rayner - agricultural policy and trade

Research Fellows (External)

V.N. Balasubramanyam (University of Lancaster) – foreign direct investment and multinationals
David Fielding (Leicester University) - investment, monetary and fiscal policy
Göte Hansson (Lund University) – trade, Ethiopian development
Stephen Knowles (University of Otago) – inequality and growth
Robert Lensink (University of Groningen) – aid, investment, macroeconomics
Scott McDonald (Sheffield University) – CGE modelling, agriculture
Mark McGillivray (RMIT University) - aid allocation, human development
Doug Nelson (Tulane University) - political economy of trade
Shelton Nicholls (University of Lancaster) - commodity prices
Eric Strobl (University College Dublin) – labour markets
Finn Tarp (University of Copenhagen) – aid, CGE modelling
Howard White (IDS) - aid, poverty